Author’s response to reviews

Title: Social capital and health during pregnancy; an in-depth exploration from rural Sri Lanka

Authors:
Thilini Agampodi (thilinichanchala@yahoo.com)
Thilde Rheinländer (thtr@sund.ku.dk)
Suneth Agampodi (sunethagampodi@yahoo.com)
Nicholas Glozier (nick.glozier@sydney.edu.au)
Sisira Siribaddana (sisira.siribaddana@gmail.com)

Version: 4 Date: 31 May 2017

Author’s response to reviews:

Dear Editor,

I have done the corrections requested. For some comments I have answered during the previous revision. Please kindly note that we had to include few sections and few reputations as a request of the previous reviewers. This might have made the article long. However we do think that their opinions for inclusion of theoretical frame work is beneficial.

Thank you very much for the constructive review and editing carried out.

Dr. T.C. Agampodi